

## 

**Citation:** Klinkenberg E, Floyd S, Shanaube K, Mureithi L, Gachie T, de Haas P, et al. (2023) Tuberculosis prevalence after 4 years of population-wide systematic TB symptom screening and universal testing and treatment for HIV in the HPTN 071 (PopART) community-randomised trial in Zambia and South Africa: A cross-sectional survey (TREATS). PLoS Med 20(9): e1004278. https://doi.org/10.1371/journal.pmed.1004278

Received: November 23, 2022

Accepted: July 27, 2023

Published: September 8, 2023

**Copyright:** © 2023 Klinkenberg et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The analysis dataset that was used to produce the results that are presented in this paper can be accessed from the LSHTM data repository, and the DOI is <a href="https://doi.org/10.17037/DATA.00003517">https://doi.org/10.17037/DATA.00003517</a>.

**Funding:** This study was funded by the European and Developing Countries Clinical Trials Partnership (EDCTP), as part of the TREATS study with grant award number RIA2016S-1632 (EK, SFI, KS, LM, TG, PdH, BK, PJD, MR, CW, JMB, NiK, NkK, RV, AS, SFi, RH, HA) HPTN 071 (PopART) RESEARCH ARTICLE

Tuberculosis prevalence after 4 years of population-wide systematic TB symptom screening and universal testing and treatment for HIV in the HPTN 071 (PopART) communityrandomised trial in Zambia and South Africa: A cross-sectional survey (TREATS)

Eveline Klinkenberg <sup>1,2,3©‡¤</sup>\*, Sian Floyd <sup>1©‡</sup>\*, Kwame Shanaube<sup>4</sup>, Linda Mureithi<sup>5</sup>, Thomas Gachie<sup>1,4</sup>, Petra de Haas<sup>3</sup>, Barry Kosloff<sup>1,4</sup>, Peter J. Dodd<sup>6</sup>, Maria Ruperez<sup>1</sup>, Chali Wapamesa<sup>4</sup>, James Michael Burnett<sup>5</sup>, Nico Kalisvaart<sup>3</sup>, Nkatya Kasese<sup>4</sup>, Redwaan Vermaak<sup>5</sup>, Albertus Schaap<sup>1,4</sup>, Sarah Fidler<sup>7</sup>, Richard Hayes<sup>1,4</sup>, Helen Ayles<sup>1,4</sup>, on behalf of the TREATS study team<sup>1</sup>

 London School of Hygiene & Tropical Medicine (LSHTM), London, United Kingdom, 2 Department of Global Health and Amsterdam Institute for Global Health and Development, Amsterdam University Medical Center, Amsterdam, the Netherlands, 3 KNCV Tuberculosis Foundation, Hague, the Netherlands,
Zambart, University of Zambia School of Public Health, Lusaka, Zambia, 5 Health Systems Trust, Cape Town, South Africa, 6 School of Health and Related Research, University of Sheffield, Sheffield, United Kingdom, 7 HIV Clinical Trials Unit, Imperial College London, London, United Kingdom

- So These authors contributed equally to this work.
- ¤ Current address: Connect TB, Hague, the Netherlands
- ‡ These authors are joint first-authors on this work.
- ¶ Membership of the TREATS Study Team is provided in the Acknowledgements.
- \* evelineklinkenberg@gmail.com (EK); sian.floyd@lshtm.ac.uk (SF)

## Abstract

## Background

Tuberculosis (TB) prevalence remains persistently high in many settings, with new or expanded interventions required to achieve substantial reductions. The HIV Prevention Trials Network (HPTN) 071 (PopART) community-randomised trial randomised 14 communities to receive the "PopART" intervention during 2014 to 2017 (7 arm A and 7 arm B communities) and 7 communities to receive standard-of-care (arm C). The intervention was delivered door-to-door by community HIV care providers (CHiPs) and included universal HIV testing, facilitated linkage to HIV care at government health clinics, and systematic TB symptom screening. The Tuberculosis Reduction through Expanded Anti-retroviral Treatment and Screening (TREATS) study aimed to measure the impact of delivering the PopART intervention on TB outcomes, in communities with high HIV and TB prevalence.

## Methods and findings

The study population of the HPTN 071 (PopART) trial included individuals aged  $\geq$ 15 years living in 21 urban and peri-urban communities in Zambia and South Africa, with a total

was sponsored by the National Institute of Allergy and Infectious Diseases (NIAID), with funding from the U.S. President's Emergency Plan for AIDS Relief (PEPFAR) (SFI, KS, BK, NkK, RV, AS, SFi, RH, HA). Additional funding (SFI, KS, BK, NkK, RV, AS, SFi, RH, HA) was provided by the International Initiative for Impact Evaluation (3ie) with support from the Bill & Melinda Gates Foundation, as well as by NIAID, the National Institute on Drug Abuse (NIDA) and the National Institute of Mental Health (NIMH), all part of the U.S. National Institutes of Health (NIH). RH, SFI, and KS were jointly funded by the UK Medical Research Council (MRC) and the UK Department for International Development (DFID) under the MRC/DFID Concordat agreement, which is also part of the EDCTP2 programme supported by the European Union. Grant Ref: MR/ R010161/1. SFi acknowledges funding from the Imperial College National Institute for Health Research Biomedical Research Centre. No funding bodies had any role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

Abbreviations: ART, antiretroviral therapy; CAB, Community Advisory Board; CHiP, community HIV care provider; CI, confidence interval; COVID-19, Coronavirus Disease 2019; DMS, data management system; HPTN, HIV Prevention Trials Network; IDP, intensive diagnostic phase; LIMS, laboratory information management system; MFS, mobile field site; MGIT, mycobacteria growth indicator tube; MTB, *M. tuberculosis*; NHLS, National Health Laboratory Service; PLWH, people living with HIV; TB, tuberculosis; TREATS, Tuberculosis Reduction through Expanded Antiretroviral Treatment and Screening; UTT, universal testing and treatment. population of approximately 1 million and an adult HIV prevalence of around 15% at the time of the trial. Two sputum samples for TB testing were provided to CHiPs by individuals who reported  $\geq$ 1 TB suggestive symptom (a cough for  $\geq$ 2 weeks, unintentional weight loss  $\geq$ 1.5 kg in the last month, or current night sweats) or that a household member was currently on TB treatment. Antiretroviral therapy (ART) was offered universally at clinics in arm A and according to local guidelines in arms B and C. The TREATS study was conducted in the same 21 communities as the HPTN 071 (PopART) trial between 2017 and 2022, and TB prevalence was a co-primary endpoint of the TREATS study. The primary comparison was between the PopART intervention (arms A and B combined) and the standard-of-care (arm C). During 2019 to 2021, a TB prevalence survey was conducted among randomly selected individuals aged >15 years (approximately 1,750 per community in arms A and B, approximately 3,500 in arm C). Participants were screened on TB symptoms and chest X-ray, with diagnostic testing using Xpert-Ultra followed by culture for individuals who screened positive. Sputum eligibility was determined by the presence of a cough for >2 weeks, or >2 of 5 "TB suggestive" symptoms (cough, weight loss for >4 weeks, night sweats, chest pain, and fever for  $\geq$ 2 weeks), or chest X-ray CAD4TBv5 score  $\geq$ 50, or no available X-ray results. TB prevalence was compared between trial arms using standard methods for cluster-randomised trials, with adjustment for age, sex, and HIV status, and multiple imputation was used for missing data on prevalent TB. Among 83,092 individuals who were eligible for the survey, 49,556 (59.6%) participated, 8,083 (16.3%) screened positive, 90.8% (7,336/8,083) provided 2 sputum samples for Xpert-Ultra testing, and 308 (4.2%) required culture confirmation. Overall, estimated TB prevalence was 0.92% (457/49,556). The geometric means of 7 community-level prevalence estimates were 0.91%, 0.70%, and 0.69% in arms A, B, and C, respectively, with no evidence of a difference comparing arms A and B combined with arm C (adjusted prevalence ratio 1.14, 95% confidence interval, CI [0.67, 1.95], p = 0.60). TB prevalence was higher among people living with HIV than HIV-negative individuals, with an age-sex-community adjusted odds ratio of 2.29 [95% CI 1.54, 3.41] in Zambian communities and 1.61 [95% CI 1.13, 2.30] in South African communities. The primary limitations are that the study was powered to detect only large reductions in TB prevalence in the intervention arm compared with standard-of-care, and the between-community variation in TB prevalence was larger than anticipated.

## Conclusions

There was no evidence that the PopART intervention reduced TB prevalence. Systematic screening for TB that is based on symptom screening alone may not be sufficient to achieve a large reduction in TB prevalence over a period of several years. Including chest X-ray screening alongside TB symptom screening could substantially increase the sensitivity of systematic screening for TB.

## **Trial registration**

The TREATS study was registered with <u>ClinicalTrials.gov</u> Identifier: <u>NCT03739736</u> on November 14, 2018. The HPTN 071 (PopART) trial was registered at <u>ClinicalTrials.gov</u> under number <u>NCT01900977</u> on July 17, 2013.

## Author summary

## Why was this study done?

- Tuberculosis (TB) prevalence remains persistently high in many settings, with new or expanded interventions required to achieve substantial reductions.
- The HIV Prevention Trials Network (HPTN) 071 Population Effects of Antiretroviral Therapy to Reduce HIV Transmission (PopART) community-randomised trial, conducted in 21 Zambian and South African communities (2013 to 2018), provided an opportunity to test whether the PopART intervention—universal testing and universal treatment for HIV combined with universal symptom screening for TB—could reduce TB prevalence.
- The Tuberculosis Reduction through Expanded Anti-retroviral Treatment and Screening (TREATS) study measured the impact of the PopART intervention on TB prevalence, comparing 14 communities that received the PopART intervention during 2014 to 2017 with 7 communities that received standard-of-care.

## What did the researchers do and find?

- We conducted a TB prevalence survey during 2019 to 2021, among around 50,000 randomly selected individuals aged ≥15 years in the 21 urban and peri-urban communities that were included in the HPTN 071 (PopART) trial.
- Overall, estimated TB prevalence was 0.92% (457/49,556). There was no evidence of a difference in TB prevalence between the PopART intervention and standard-of-care communities, with an adjusted prevalence ratio of 1.14, 95% confidence interval (CI) [0.67, 1.95].
- TB prevalence was around 2 times higher among HIV–positive than HIV–negative individuals, adjusted for age, sex, and community, and several times higher among men than women.

## What do these findings mean?

- This study found no evidence that the PopART intervention reduced TB prevalence, compared with standard-of-care.
- Systematic screening for TB that is based on symptom screening alone may not be sufficient to achieve a large reduction in TB prevalence over a period of a few years.
- Including chest X-ray screening alongside TB symptom screening could substantially increase the sensitivity of systematic screening for TB, and several years of delivering such a strategy might be able to achieve a considerable reduction in TB prevalence.

## Introduction

Tuberculosis (TB) overtook HIV as the leading infectious cause of death worldwide, prior to the Coronavirus Disease 2019 (COVID-19) pandemic, requiring a major policy shift if it is to be controlled in line with the global commitment to "end TB" [1]. One potential strategy, as outlined in WHO updated screening guidelines of 2021, is to increase systematic screening for TB in settings with high TB prevalence (>0.5%), with diagnostic testing offered to individuals who screen positive (for example on TB symptoms, chest X-ray, or belonging to a "high-risk" population such as household contacts or HIV–positive individuals) [2] or to the whole population [3]. This would have an immediate effect on TB prevalence, and by facilitating early diagnosis has the potential to reduce transmission of infection and later TB incidence and prevalence. Systematic screening for HIV–positive individuals has been recommended by WHO since 2011 [4]; in 2015, WHO guidelines [5] also recommended that antiretroviral therapy (ART) should be offered universally to HIV–positive individuals who know their HIV–positive status, with a reduction in TB incidence among such individuals being one of the expected health benefits.

Meanwhile, 4 community-randomised trials have shown that delivery of universal HIV testing, and universal ART for HIV–positive individuals (universal testing and treatment, UTT), achieves high ART coverage and viral suppression among HIV–positive individuals [6,7]. Two of the trials also found evidence that delivery of UTT reduces HIV incidence in the general adult population. It is possible that delivery of UTT also reduces TB incidence among HIV– positive individuals (through increased ART coverage), and following this it may indirectly reduce general-population TB incidence and prevalence [8].

The HIV Prevention Trials Network (HPTN) 071 (PopART) community-randomised trial was conducted during 2013 to 2018 in 21 communities in Zambia and the Western Cape Province of South Africa and was designed to measure the impact of a combination HIV prevention package (the PopART intervention) on HIV incidence [9]. The study built on previous TB and HIV research that was conducted in many of the same communities, including the ZAMSTAR trial during 2004 to 2011 [10,11]. The PopART package of interventions was centred on universal HIV testing and support for linkage to HIV care, with or without universal ART, and included systematic screening for TB based on TB symptoms. During 2015 to 2018 HIV incidence was on average around 20% lower across the 14 communities that were randomised to the standard-of-care arm [7].

To contribute to the evidence base on how to reduce TB burden in HIV-endemic settings, the TREATS (Tuberculosis Reduction through Expanded Anti-retroviral Treatment and Screening) study was designed to measure the population-level impact of the PopART intervention on TB prevalence, transmission, and TB notifications. We report here on the impact of the PopART intervention on TB prevalence; this was measured during 2019 to 2021, 2 to 4 years after the last year of the PopART intervention delivery.

### **Methods**

#### Ethics statement

The TREATS study was approved by the research ethics committees of the London School of Hygiene & Tropical Medicine, the University of Zambia (UNZABREC), and Pharma-Ethics (Pty) in South Africa. Individuals gave written informed consent to survey participation, and for individuals aged 15 to 17 years, assent and parental consent were obtained.

The TREATS study was registered with <u>ClinicalTrials.gov</u> Identifier: NCT03739736 on November 14, 2018. The HPTN 071 (PopART) trial was registered at <u>ClinicalTrials.gov</u> under number NCT01900977 on July 17, 2013.

# HPTN 071 (PopART) community-randomised trial design, PopART intervention, and TREATS study design

The HPTN 071 (PopART) community-randomised trial was conducted during 2013 to 2018. All communities were urban or peri-urban, 12 were in Zambia and 9 were in the Western Cape Province of South Africa, with a total population of around 1 million individuals and estimated adult HIV prevalence of around 15%. In the HPTN 071 (PopART) trial, the 21 communities were matched into 7 triplets based on geographical area and estimated HIV prevalence [9]. Within each triplet, 1 community was randomised to arm A, 1 to arm B, and 1 to arm C, with restricted randomisation used to ensure overall balance among the 3 trial arms in terms of HIV prevalence, ART coverage, and population size. The randomisation of communities to trial arms was done in a public randomisation ceremony in 2013 [9]. The 3 trial arms were arm A (PopART intervention, with UTT and also systematic screening for TB based on TB symptoms), arm B (as arm A, except ART according to local guidelines [universal ART from mid-2016]), or arm C (standard-of-care).

The PopART intervention was delivered by community HIV care providers (CHiPs) for 4 years (2014 to 2017), primarily through a door-to-door approach, with CHiPs aiming to reach all households and all community members who were aged  $\geq$ 15 years approximately annually. HIV testing was offered to all who participated in the intervention, apart from individuals who self-reported they were HIV-positive. Systematic screening for TB was done for everyone who reported they were not currently on TB treatment, based on TB symptoms. Two sputum samples were requested from individuals who reported any among a cough for  $\geq 2$  weeks, unintentional weight loss > 1.5 kg in the last month, current night sweats, or that a member of their household was currently on TB treatment. CHiPs took the sputum samples to the health clinic, where TB diagnostic testing followed national guidelines; in South African communities Xpert-MTB/RIF was used, and in Zambian communities Xpert-MTB/RIF was used for people living with HIV (PLWH) and smear microscopy was used for all other individuals. CHiPs made at least 1 follow-up visit to individuals who had TB symptoms or were a household contact of a TB patient at the time of the annual visit, to provide TB diagnostic test results and support for starting TB treatment (if required). They also made regular follow-up visits to individuals who were known to be HIV-positive, to support linkage to HIV care and retention on ART. Alongside this, routine TB service provision at the health clinic was strengthened in various ways, in collaboration with each country's Department of Health or Ministry of Health. This strengthening included additional provision of machines for Xpert-MTB/RIF testing, additional Xpert-MTB/RIF cartridges and reagents, support to staff doing laboratory testing, and support to strengthen recordkeeping for TB registers. TB preventive therapy was recommended for PLWH and individuals aged <5 years old who were a household contact of a TB patient, according to national guidelines, but uptake of such therapy was low during the period of the PopART intervention delivery.

The TREATS study was conducted in the same 21 communities as the HPTN 071 (PopART) trial and was designed as a follow-up study to the HPTN 071 (PopART) trial. The TREATS study was designed to measure the impact of the PopART intervention on TB outcomes, including TB prevalence, and was conducted during 2017 to 2022. The primary comparison of TB prevalence was between arms A and B combined (PopART intervention) versus arm C (standard-of-care). Arms A and B were combined because the TB screening and

universal HIV testing components of the PopART intervention were the same throughout the HPTN 071 (PopART) trial, and universal ART was offered in arm B from mid-2016 (when universal ART became national guidelines).

## **TREATS TB prevalence survey design**

Sample size and survey timing. The TREATS TB prevalence survey was conducted among a random sample of individuals aged  $\geq$ 15 years during 2019 to 2021. The target sample size was 3,000 to 4,000 participants in each arm C community and 1,500 to 2,000 participants in each arm A and arm B community, with a total targeted sample size of around 50,000 participants. With an average of 1,750 participants in each arm A and arm B community, an average of 3,500 participants in each arm C community, an assumed coefficient of variation k of 0.25, and an assumed 50% reduction in TB prevalence in arms A and B compared with arm C, study power was 91% if average arm C TB prevalence was 1%, and 89% if average arm C TB prevalence was 0.8%. The corresponding figures for study power to show a 45% reduction in TB prevalence were 84% and 81%, respectively. The assumption that TB prevalence could be reduced by around 50% after several years of delivery of the PopART intervention was informed by mathematical modelling projections that were done before the start of the TREATS study (see <u>S1 Protocol</u> for more details).

Community sensitisation was conducted prior to the start of prevalence survey activities, including distribution of information leaflets, door-to-door information provision, and involvement of the Community Advisory Board (CAB).

**Field procedures.** Within each community, random sampling was structured according to geographically defined blocks of around 200 households. For every randomly selected block, all households were visited by a research assistant. If an adult household member was found at home, permission was sought to enumerate (list) all household members. In enumerated households, an individual was eligible to participate if they were a community resident aged  $\geq$ 15 years. Eligible individuals were given barcoded invitation cards and invited to attend a mobile field site (MFS) that was located within the community as close to the sampled block as possible. A OneStopTB Platform (a truck, containing a digital X-ray and mobile laboratory fitted with an Xpert instrument) was stationed at the MFS, and several tents ("stations") were set up to provide spaces to administer questionnaires and collect samples.

All individuals who attended the MFS and consented to participate in the survey followed a defined order of procedures. First, questionnaire information including sociodemographic and socioeconomic characteristics, TB symptoms, previous history of TB treatment, and behavioural risk factors for TB (smoking, drinking alcohol, occupational exposure), was collected. Subsequently, individuals were invited to have a digital chest X-ray taken, with images read using computer-aided detection software (CAD4TB, version 5.0, Delft Imaging, the Netherlands) that provided a score between 0% and 100% representing the probability that an individual has TB. After this, self-reported information on previous HIV testing and current and past ART use was collected, including whether an individual had ever tested for HIV, the date and result of their last test, and whether they had tested for HIV in the previous 12 months. Following this, HIV testing was offered to all individuals who did not self-report they were HIV-positive. All individuals who tested HIV-positive were referred to HIV services following counselling. An individual was considered to have participated in the survey if they were screened for TB symptoms and attended the X-ray station.

Eligibility to provide 2 sputum samples (S1 and S2) for Xpert-Ultra testing. "Sputumeligible" individuals were identified based on TB screening using symptoms and chest X-ray. An individual was sputum-eligible based on TB symptoms if they had a cough for  $\geq$ 2 weeks or if they had 2 or more among 5 "TB suggestive" symptoms (cough of any duration, unexpected weight loss for  $\geq$ 4 weeks, night sweats for  $\geq$ 2 weeks, chest pains for  $\geq$ 2 weeks, fever for  $\geq$ 2 weeks). An individual was sputum-eligible based on chest X-ray if their X-ray CAD4TBv5 score was  $\geq$ 50, or they did not have an X-ray done, or the X-ray was done but the X-ray score was not captured (S1 Fig). All sputum-eligible individuals were requested to provide 2 "on-the-spot" sputum samples (S1 and S2, taken  $\geq$ 30 min apart), for Xpert-Ultra testing within the next 24 h; this testing was done in the OneStopTB truck, and followed the manufacturer's standard operating procedures [12]. Sputum-eligible individuals were asked to return the following day to receive their Xpert-Ultra test results, and during this "day 2" visit they were also reviewed by a medical officer who made decisions on referral to TB or other care based on their screening and test results. There was active tracing of individuals who did not return for the "day 2" follow-up, with a focus on tracing those with positive Xpert-Ultra test results.

Xpert-Ultra results for sputum samples S1 and S2 were classified using the test read-out as *M. tuberculosis* (MTB) not detected, MTB-trace-detected, MTB detected very low, low, medium, or high; or as non-determinate (invalid, error, or no result). If the Xpert-Ultra test result was "non-determinate," the test was repeated up to 2 times, either on the original sample or (if needed) on a new sample.

**Eligibility to provide sputum samples (S3, or S3 and S4) for culture testing.** Collection of sputum samples for culture testing, and the number of samples requested, differed between the first 4 study communities (in which an "intensive diagnostic phase" (IDP) was conducted) and the remaining 17 "non-IDP" communities.

In the 4 IDP communities, all sputum-eligible individuals who returned on "day 2" were requested to provide 1 sputum sample (S3) for culture testing. Comparison of the Xpert-Ultra and culture results in these IDP communities was used to define a subgroup of sputum-eligible individuals who would be "culture eligible" in the 17 non-IDP communities [13].

In the 17 non-IDP communities, sputum-eligible participants who had valid Xpert-Ultra test results from both of the S1 and S2 sputum samples were classified into one of 3 categories (S2 and S3 Figs). First, if both of the S1 and S2 Xpert results were "*M. tuberculosis* (MTB) not detected," then the individual was not culture-eligible and was considered not to have prevalent TB. Second, if both of the S1 and S2 Xpert results were "MTB detected" with a grade of very low or above, and at least one of them was graded as low, medium, or high, then the individuals with other combinations of S1 and S2 Xpert test results (for example, both samples "MTB detected" and 1 sample "MTB detected") were "culture-eligible." Culture-eligible individuals were requested to provide 2 additional sputum samples (S3 and S4) for culture testing, and these culture test results were used to determine their prevalent TB status.

**Culture testing of sputum samples.** All sputum samples that were collected for culture testing were batched and kept in a refrigerator until being transported on the same day to the central laboratory in a cooler box. In Zambia, samples were taken to the Zambart central laboratory in Lusaka, and in South Africa to the National Health Laboratory Service (NHLS) laboratory in Greenpoint, Cape Town. At the culture laboratory, culture testing followed standard procedures that have been described previously [13] and further details are given also in <u>S1</u> Text and <u>S4 Fig</u>. In short, each sputum sample was inoculated onto 2 mycobacteria growth indicator tubes (MGITs) and incubated for 42 days or until growth was observed. Diagnostic testing was done on growth-positive tubes (<u>S4 Fig</u>, step 2).

The culture result for each sputum sample was defined based on the combination of the culture results from the 2 tubes (S4 Fig, steps 3 to 5). A sample was classified as culture-positive for *M. tuberculosis* if  $\geq$ 1 tube result was positive for *M. tuberculosis*. Among samples that were

not culture-positive for *M. tuberculosis*, samples were classified as culture-negative if  $\geq 1$  tube was culture-negative or  $\geq 1$  tube was positive for non-tuberculous-mycobacteria. The result was classified as contaminated if both tubes were contaminated and as non-interpretable if either both tubes were non-interpretable, or one was non-interpretable and one was contaminated. We defined a sputum sample culture result as valid if it was culture-positive for *M. tuberculosis* or if it was culture-negative, and also from a batch where the positive control grew and the negative control did not; other results were classified as missing.

The culture result for each individual was defined based on the combination of the culture results from the S3 and S4 samples. An individual had a valid culture result if one or both of the S3 and S4 sample results were valid and was classified as having prevalent TB if  $\geq 1$  sample was culture-positive for *M. tuberculosis*, and as not having prevalent TB if neither sample was culture-positive for *M. tuberculosis* and  $\geq 1$  sample was culture-negative, with other individuals classified as having missing data on their culture status (S4 Fig, step 5).

**Definition of the primary outcome of prevalent TB.** Sputum-eligible participants who had valid Xpert-Ultra results from both S1 and S2 and were not culture-eligible (according to the non-IDP community definition) were classified as prevalent TB (yes or no) based solely on their Xpert-Ultra results (see above).

Sputum-eligible participants who were culture-eligible (according to the non-IDP community definition) were classified as prevalent TB (yes or no) based solely on their culture results. Those who were culture-positive for *M. tuberculosis* were considered to have prevalent TB, those who were culture-negative for *M. tuberculosis* were considered not to have prevalent TB, and those with no valid culture result were classified as having missing data on prevalent TB.

Sputum-eligible participants with missing Xpert-Ultra data on one or both of the S1 and S2 sputum samples (a test result that was non-determinate or a sample that was not provided) were classified as having missing data on prevalent TB. It was assumed that participants who were not eligible to provide sputum samples did not have prevalent TB.

**Data capture and statistical analysis.** Data were captured digitally into an electronic data management system (DMS) specifically designed for the TREATS TB prevalence survey while culture data were captured using each laboratory's existing laboratory information management system (LIMS).

For sputum-eligible participants whose prevalent TB status could not be determined due to missing Xpert-Ultra or culture test results, imputation of the missing data was done (further details in S2 Text). The first step was to impute missing data on prevalent TB (yes or no) among culture-eligible individuals with missing data on culture results, and the second step was to impute missing data on prevalent TB (yes or no) for the sputum-eligible individuals who had missing Xpert-Ultra results. A total of 300 imputed datasets were created (30 times 10; 30 imputed datasets for step 1 and for each of these 30 datasets, 10 imputations were done for step 2).

TB prevalence was then compared among trial arms using a standard "two-stage" approach for the analysis of cluster-randomised trials with <15 clusters per trial arm [14], but accounting for the missing data by applying this two-stage approach to each of the 300 imputed datasets. In Stage 1, a logistic regression model was fitted with prevalent TB (yes or no) as the outcome and study triplet, age group, sex, and HIV status as explanatory variables; one regression model was fitted for all Zambian communities (4 study triplets and 12 communities, with study triplet as a "fixed" effect) and one regression model was fitted for all South African communities (3 triplets and 9 communities, with study triplet as a "fixed" effect). The adjustment for study triplet, age group, sex, and HIV status was prespecified in the statistical analysis plan (see <u>Supporting information file S1 SAP</u>). The country-specific logistic regression models were then used to predict the probability that an individual had prevalent TB, under the null hypothesis of no intervention effect. These predicted probabilities were then summed across all individuals in the same community, as an estimate of the expected number (E) of individuals in the community with prevalent TB under the null hypothesis. In subgroup analysis, the fitting of the Stage 1 model and the summation of observed and expected values were done with restriction to individuals belonging to the subgroup—for example with restriction to males, or with restriction to HIV-negative individuals. In Stage 2, for each community the ratio of the observed (including individuals imputed to have prevalent TB) number of individuals with prevalent TB (O) to the expected number (E) was calculated. A linear regression model with log(O/E) as the outcome and triplet and trial arm as explanatory variables was then fitted, to obtain estimates of the TB prevalence ratios (and corresponding 95% confidence intervals (CIs)) comparing arms A and B combined with arm C (primary analysis) as well as comparing each of arms A and B separately with arm C. Lastly, the prevalence ratio estimates from each of the imputed datasets were averaged (using a geometric mean) to obtain an overall summary estimate, and Rubin's rules were used to obtain 95% CIs that accounted for the imputation of missing data. Subgroup analysis for males and females, older and younger individuals, HIV-positive and HIV-negative individuals, and individuals who had been resident in the study communities for  $\geq$ 5 years, was prespecified. Analyses were conducted in Stata 17.

## Results

### **Enumeration and participation**

A total of 122,381 household members (all ages) were enumerated, 89,850 in Zambia and 32,531 in South Africa, 83,092 were eligible to participate, and 49,556 (59.6%) participated (Figs 1 and <u>\$5</u>). The participation rate varied slightly by trial arm (62.2% in arm A, 60.9% in arm B, 57.7% in arm C, S6 Fig), was higher in South African than Zambian communities (69.3% versus 55.0%, S5 Fig), and higher in females than males (66.1% versus 51.8%). The proportion of participants who were eligible to provide sputum samples based on the TB symptoms they reported was similar in South African (4.4%) and Zambian (3.8%) communities and varied slightly by trial arm (4.7%, 4.3%, and 3.6% in arms A, B, and C, respectively). The proportion of participants who were eligible to provide sputum samples based on an abnormal Xray was higher in South African (16.8%) than Zambian (11.4%) communities, and similar by trial arm (13.4%, 13.4%, and 13.0% in arms A, B, and C, respectively). Overall, 8,083 (16.3%) participants were sputum-eligible and this was similar among trial arms (16.7%, 17.3%, and 15.6% in arms A, B, and C, respectively, Figs 1 and S6), and 80.5% (6,508/8,083) returned for the day 2 visit (78.9% in arm A, 79.8% in arm B, and 81.9% in arm C). Day 2 attendance differed by country, being 77% in Zambian communities and 84% for South African communities (S5 Fig).

## Characteristics of participants

The age-sex profile of participants was similar by trial arm (Table 1). A slightly higher proportion of participants had been resident in the community for  $\geq$ 5 years in arms A and B than in arm C (73.7% in arms A and B combined versus 69.9% in arm C). Education level patterns were similar by trial arm, though in arm C a slightly higher proportion had completed secondary school grade or higher education (72.2% in arms A and B combined versus 77.3% in arm C).

Overall, 4,356 (8.8%) of participants self-reported they had previously been treated for TB and 295 (0.4%) self-reported being on TB treatment on the day of the survey. This was higher in South Africa (14.2% and 1.1%, respectively) than in Zambia (5.5% and 0.3%, respectively) (S1 Table). Self-reported previous TB treatment and current TB treatment were slightly higher



**Fig 1. Flow diagram by trial arm of participant enrolment, sample collection, and MTB detection.** MTB, *M. tuberculosis*; TB, tuberculosis; UTT, universal testing and treatment.

https://doi.org/10.1371/journal.pmed.1004278.g001

in arm A (9.4% and 0.9%, respectively) than in arm B (8.3% and 0.6%) and arm C (8.7% and 0.4%).

Overall, 96% (47,581/49,556) of participants completed the HIV interview, and this did not differ by trial arm (S2 Table). Overall, 42,008 (88.3%) reported they had previously been tested for HIV, and this was slightly lower in arm C (85.9%) than in arm A (91.4%) and arm B (89.5%). Overall, 68.2% either self-reported they were HIV–positive or that they had tested

	Arı	n A	Arm B		Arm C		Totals	
	n	%	n	%	n	%	n	%
Number seen at site (fully screened)	13,906		11,787		23,863		49,556	
Number female	8,552	61.5%	7,282	61.8%	14,252	59.7%	30,086	60.7%
Age group								
15-24 years	5,056	36.4%	4,417	37.5%	9,114	38.2%	18,587	37.5%
25-34 years	3,639	26.2%	2,930	24.9%	6,073	25.4%	12,642	25.5%
35-44 years	2,357	16.9%	1,995	16.9%	3,866	16.2%	8,218	16.6%
45–54 years	1,418	10.2%	1,205	10.2%	2,288	9.6%	4,911	9.9%
55+ years	1,436	10.3%	1,240	10.5%	2,522	10.6%	5,198	10.5%
Length of residence								
<1 years	685	4.9%	570	4.8%	1,583	6.6%	2,838	5.7%
1–5 years	2,984	21.5%	2,528	21.4%	5,594	23.4%	11,106	22.4%
>5-10 years	3,226	23.2%	2,909	24.7%	4,900	20.5%	11,035	22.3%
>10 years	7,011	50.4%	5,780	49.0%	11,786	49.4%	24,577	49.6%
Education level								
None	483	3.5%	502	4.3%	626	2.6%	1,611	3.3%
Primary school	3,412	24.5%	2,750	23.3%	4,791	20.1%	10,953	22.1%
Secondary school	9,405	67.6%	8,060	68.4%	17,276	72.4%	34,741	70.1%
Higher education	606	4.4%	475	4.0%	1,170	4.9%	2,251	4.5%
History TB treatment								
Previous TB	1,304	9.4%	976	8.3%	2,076	8.7%	4,356	8.8%
Current TB	128	0.9%	76	0.6%	91	0.4%	295	0.6%
HIV status*								
Negative	10,688	76.9%	8,916	75.6%	17,391	72.9%	36,995	74.7%
Positive	2,322	16.7%	1,847	15.7%	3,808	16.0%	7,977	16.1%
Unknown	896	6.4%	1,024	8.7%	2,664	11.2%	4,584	9.3%
HIV and ART status								
Negative	10,688	76.9%	8,916	75.6%	17,391	72.9%	36,995	74.7%
Self-reported positive, on ART	1,962	14.1%	1,476	12.5%	3,160	13.2%	6,598	13.3%
Self-reported positive, not on ART	151	1.1%	133	1.1%	285	1.2%	569	1.1%
Tested HIV-positive in survey	209	1.5%	238	2.0%	363	1.5%	810	1.6%
Unknown	896	6.4%	1,024	8.7%	2,664	11.2%	4,584	9.3%

#### Table 1. Characteristics (absolute numbers and relevant proportions) of the study population by arm and overall combined.

\* After completing TREATS study procedures so including testing done as part of TREATS.

ART, antiretroviral therapy; HIV, human immunodeficiency virus; TB, tuberculosis; TREATS, Tuberculosis Reduction through Expanded Anti-retroviral Treatment and Screening.

https://doi.org/10.1371/journal.pmed.1004278.t001

HIV–negative in the previous 12 months, and this was slightly higher in arm A (71.4%) than in arm B (67.0%) and arm C (67.0%). Overall, 14.5% (7,181/49,556) of participants selfreported they were HIV–positive, and among them 92.1% (6,613/7,181) self-reported they were currently on ART, and these figures were similar by trial arm.

Among the 40,409 participants who completed the HIV interview and did not self-report being HIV–positive, 73.9% (n = 29,882) accepted an offer of HIV testing. The proportion who accepted this offer was slightly lower in arm C (70.2%) than in arm A (74.6%) and arm B (81.0%), and 2.7% (n = 803) tested HIV–positive and this was similar by trial arm (S2 Table).

Following survey participation, 16.1% of participants were known to be HIV–positive; 13.3% of participants self-reported they were HIV–positive and taking ART, 1.1% self-reported

they were HIV–positive and not taking ART, and 1.6% did not self-report as HIV–positive but tested HIV–positive in the survey. Seventy-five percent of participants tested HIV–negative as part of the survey or self-reported an HIV–negative test in the previous 12 months, and the HIV status of the remaining 9.3% was considered unknown. Figures were similar by trial arm, apart from a slightly higher percentage being of unknown HIV status in arm C (Table 1 and S2 Table).

Supporting information <u>S5</u> and <u>S6</u> Figs and <u>S1–S3</u> Tables provide further details by country and arm.

# Provision of sputum samples, valid Xpert-Ultra results, culture eligibility, and valid culture results

Around 93% of sputum-eligible individuals (7,502/8,083) produced at least 1 sputum sample for Xpert-Ultra testing, 91% provided 2 samples (S1 and S2), and 88% (7,138/8,083) had valid Xpert-Ultra results from both sputum samples, among whom 4.3% (308/7,138) were cultureeligible (Fig 1). The proportions were similar by trial arm. Among the 308 culture-eligible individuals, 91% (279/308) attended the day 2 visit, for 227/279 (81%) an S3 sample was provided and for 168 (60%) an S4 sample was provided; 198 (64%) of culture-eligible individuals had valid culture results and this proportion was higher in arm A than in arms B and C (76%, 57%, and 61% in arms A, B, and C, respectively) (Figs 1 and S6). Overall, 352 participants were diagnosed with prevalent TB, 129 in Zambia and 223 in South Africa. Twenty-eight percent of individuals with prevalent TB screened positive on TB symptoms (42% in Zambian and 20% in South African communities, S3 Table).

#### TB prevalence estimates and comparison among trial arms

Overall, the estimated proportion of survey participants with prevalent TB (after imputing missing data) was 0.92% (457/49,556), lower in Zambian (0.51%, 158/30,908) than South African communities (1.60%, 299/18,648). TB prevalence estimates varied substantially among communities (Table 2), in Zambia ranging from 0.13% in the arm B and arm C study communities in triplet 4 to 0.84% in the arm A community in triplet 2 and the arm B community in triplet 3, and in South Africa ranging from 1.11% in the arm A community in triplet 5 to 2.33% in the arm C community in triplet 7. The coefficient of between-community variation k was approximately 0.3, after accounting for between-triplet and between-arm variation.

Across the communities in arms A and B, the geometric mean of the 14 community-level TB prevalences was 0.80%, with figures of 0.91% for arm A and 0.70% for arm B (<u>Table 2</u>). Across the arm C communities, the geometric mean of the 7 community-level TB prevalence values was 0.69% (<u>Table 2</u>).

The adjusted prevalence ratio comparing the combined intervention arm (arms A and B) versus the standard-of-care arm (arm C) was 1.14 (95% CI 0.67, 1.95) (Table 2), with no evidence of a difference between the trial arms (p = 0.60). There was also no evidence of a difference in TB prevalence between arm A and arm C communities, or between arm B and arm C communities.

TB prevalence was higher in males than females (for example, the geometric mean of the 14 community-level TB prevalence values in arms A and B combined was 1.46% among males compared with 0.36% among females), among older ( $\geq$ 30 years) compared with younger (<30 years) individuals (geometric means of 1.15% and 0.33%, respectively, in arms A and B combined) and among HIV-positive compared with HIV-negative individuals (geometric means of 1.13% and 0.64%, respectively, in arms A and B combined) (Table 2). Across these subgroups, there was no evidence that TB prevalence was different in arms A and B combined

Table 2.	TB prevalence by	v trial arm, overa	ll and stratified by	y sex, age group,	and HIV status.
----------	------------------	--------------------	----------------------	-------------------	-----------------

		Arm A		Arn	n B	Arm A+B		Arm C	
Overall	Triplet	$n/N^1$	% <sup>2</sup>	$n/N^1$	% <sup>2</sup>	$n/N^1$	% <sup>2</sup>	$n/N^1$	% <sup>2</sup>
	1	11.4/1,971	0.58	12.9/1,905	0.67	24.3/3,876	0.62	10.4/3,494	0.30
	2	17.5/2,088	0.84	4.9/2,122	0.23	22.4/4,210	0.52	22.6/4,064	0.56
	3	20.7/2,887	0.72	18.7/2,235	0.84	39.4/5,122	0.78	22.1/3,813	0.58
	4	10.9/1,609	0.68	2.1/1,548	0.13	13.0/3,157	0.41	4.2/3,172	0.13
	5	22.7/2,049	1.11	24.5/1,804	1.36	47.2/3,853	1.25	51.9/4,022	1.29
	6	20.7/1,772	1.17	15.5/693	2.24	36.2/2,465	1.46	61.6/3,042	2.03
	7	25.5/1,530	1.66	23.9/1,480	1.62	49.4/3,010	1.63	52.7/2,256	2.33
	Overall (triplets 1–7 combined)	129.4/13,906	0.93	102.4/11,787	0.87	231.8/25,693	0.90	225.5/23,863	0.94
	TB prevalence, geometric mean <sup>3</sup>		0.91		0.70		0.80		0.69
	Unadjusted PR, 95% CI, <i>p</i> -value	1.31	[0.70, 2.45] p = 0.36	1.01	[0.54, 1.90] p = 0.97	1.15	[0.67, 1.98] p = 0.58	referent	
	aPR <sup>4</sup> , 95% CI, <i>p</i> -value	1.29	[0.69, 2.39] p = 0.38	1.01	[0.54, 1.88] <i>p</i> = 0.97	1.14	[0.67, 1.95] <i>p</i> = 0.60	referent	
Males	Overall (triplets 1–7 combined)	93.7/5,354	1.75	71.5/4,505	1.59	165.2/9,859	1.68	139.0/9,611	1.45
	TB prevalence, geometric mean		1.70		1.25		1.46		1.07
	aPR, 95% CI, <i>p</i> -value	1.53	[0.79, 2.95] p = 0.18	1.12	[0.58, 2.18] p = 0.70	1.31	[0.74, 2.32] p = 0.32	referent	
Females	Overall (triplets 1–7 combined)	35.7/8,552	0.42	30.9/7,282	0.42	66.6/15,834	0.42	86.5/14,252	0.61
	TB prevalence, geometric mean		0.39		0.33		0.36		0.42
	aPR, 95% CI, <i>p</i> -value	0.90	[0.40, 2.04] p = 0.78	0.80	[0.35, 1.80] p = 0.55	0.85	[0.42, 1.71] <i>p</i> = 0.61	referent	
Age <30 years	Overall (triplets 1–7 combined)	45.0/7,094	0.63	28.4/6,095	0.47	73.4/13,189	0.56	70.2/12,517	0.56
	TB prevalence, geometric mean		0.60		0.18		0.33		0.36
	aPR, 95% CI, <i>p</i> -value	1.73	[0.66, 4.54] p = 0.23	0.86	[0.32, 2.32] p = 0.74	1.22	[0.53, 2.84] <i>p</i> = 0.60	referent	
Age $\geq$ 30 years	Overall (triplets 1–7 combined)	84.4/6,812	1.24	74.0/5,692	1.30	158.4/12,504	1.27	155.3/11,346	1.37
	TB prevalence, geometric mean		1.21		1.09		1.15		1.04
	aPR, 95% CI, <i>p</i> -value	1.16	[0.68, 1.98] p = 0.56	1.05	[0.62, 1.79] p = 0.84	1.10	[0.69, 1.75] <i>p</i> = 0.65	referent	
HIV-negative	Overall (triplets 1-7 combined)	81.7/10,688	0.76	64.9/8,916	0.73	146.6/19,604	0.75	148.0/17,391	0.85
	TB prevalence, geometric mean		0.73		0.57		0.64		0.59
	aPR, 95% CI, <i>p</i> -value	1.25	[0.65, 2.42] p = 0.47	0.98	[0.51, 1.91] <i>p</i> = 0.96	1.11	[0.62, 1.97] <i>p</i> = 0.70	referent	
HIV-positive	Overall (triplets 1-7 combined)	33.2/2,322	1.43	22.7/1,847	1.23	55.9/4,169	1.34	53.7/3,808	1.41
	TB prevalence, geometric mean		1.28		1.00		1.13		1.13
	aPR, 95% CI, <i>p</i> -value	1.16	[0.46, 2.89] p = 0.72	0.97	[0.38, 2.46] p = 0.94	1.06	[0.48, 2.35] p = 0.87	referent	

<sup>1</sup> Number of individuals with prevalent TB/total participants. Number of individuals with prevalent TB is calculated after multiple imputation of missing data on prevalent TB status and is an average across 300 imputed datasets—hence not a whole number.

<sup>2</sup> Percentage with prevalent TB.

<sup>3</sup> Geometric mean of TB prevalence across 7 communities.

<sup>4</sup> Adjusted triplet, sex, age group, HIV status.

aPR, adjusted PR; CI, confidence interval; HIV, human immunodeficiency virus; PR, prevalence ratio; TB, tuberculosis.

https://doi.org/10.1371/journal.pmed.1004278.t002

than in Arm C, for example among males the adjusted prevalence ratio was 1.31 [95% CI 0.74, 2.32] (Table 2). There was also no evidence of a difference between arms A and B combined compared with arm C when analysis was restricted to individuals who had been resident in the study communities for  $\geq$ 5 years (adjusted prevalence ratio 1.26, 95% CI 0.68, 2.34).

## TB prevalence estimates, by country and HIV status

TB prevalence was higher among HIV–positive than HIV–negative individuals across almost all combinations of country, sex, and age group (Table 3). In Zambian communities, the age-adjusted odds ratios comparing HIV–positive to HIV–negative individuals were 1.80 [95% CI 1.10, 2.94] among males and 3.75 [95% CI 1.84, 7.66] among females, with weak evidence that the odds ratios differed between males and females (p = 0.09). In South African communities, the age-adjusted

	HIV-negative			HIV-positive <sup>1</sup>			Unadjusted OR, comparing HIV–positive with HIV–negative individuals	Adjusted OR <sup>2</sup> , 95% CI	
Zambia	n <sup>3</sup>	N <sup>4</sup>	% <sup>5</sup>	n <sup>3</sup>	N <sup>4</sup>	% <sup>5</sup>	OR	OR	95% CI
Male, age <30	28.0	5,886	0.48	1.7	162	1.05	2.04	2.06	[0.32, 13.42]
Male, age 30–39	18.7	1,590	1.17	12.8	339	3.78	3.31	3.44	[1.62, 7.30]
Male, age 40–49	20.0	877	2.28	12.6	447	2.81	1.24	1.24	[0.59, 2.61]
Male, age $\geq$ 50	9.4	1,063	0.89	2.1	333	0.64	0.72	0.77	[0.17, 3.59]
Male, all ages	76.1	9,416	0.81	29.2	1,281	2.28	2.86	1.80	[1.10, 2.94]
									$P = 0.19^{6}$
Female, age <30	9.6	8,857	0.11	3.4	842	0.41	3.67	3.74	[0.91, 15.37]
Female, age 30-39	1.8	2,579	0.07	9.6	1,294	0.74	11.45	11.71	[1.72, 79.90]
Female, age 40-49	1.8	1,163	0.16	6.4	1,042	0.62	4.13	4.33	[0.66, 28.47]
Female, age $\geq$ 50	6.8	1,617	0.42	2.5	509	0.49	1.15	1.20	[0.25, 5.73]
Female, all ages	20.0	14,216	0.14	22.0	3,687	0.60	4.25	3.75	[1.84, 7.66]
									$P = 0.34^{6}$
Males and females, all ages	96.1	23,632	0.41	51.2	4,968	1.03	2.55	2.29	[1.54, 3.41]
South Africa									
Male, age <30	44.8	3,114	1.44	2.6	92	2.88	1.99	1.97	[0.51, 7.64]
Male, age 30–39	28.5	1,181	2.41	6.4	222	2.88	1.18	1.26	[0.48, 3.31]
Male, age 40–49	31.1	675	4.60	8.7	203	4.28	0.92	1.02	[0.43, 2.39]
Male, age $\geq$ 50	31.1	944	3.29	7.8	141	5.56	1.72	1.98	[0.82, 4.76]
Male, all ages	135.5	5,914	2.29	25.6	658	3.89	1.72	1.40	[0.85, 2.29]
									$P = 0.70^{6}$
Female, age <30	28.8	3,662	0.79	10.1	549	1.83	2.35	2.58	[1.17, 5.71]
Female, age 30-39	18.0	1,468	1.22	13.8	950	1.46	1.19	1.42	[0.64, 3.14]
Female, age 40-49	3.6	867	0.42	7.5	586	1.28	3.19	3.92	[0.90, 17.02]
Female, age $\geq$ 50	12.6	1,452	0.87	1.5	266	0.56	0.60	0.76	[0.10, 5.60]
Female, all ages	63.0	7,449	0.85	32.9	2,351	1.40	1.66	1.93	[1.14, 3.25]
									$P = 0.40^{6}$
Males and females, all ages	198.5	13,363	1.49	58.4	3,009	1.94	1.31	1.61	[1.13, 2.30]

#### Table 3. TB prevalence by country and HIV status, overall and stratified by sex and age group.

<sup>1</sup> In Zambia: Overall, 13.5% self-reported they were HIV–positive and taking ART, 1.0% self-reported they were HIV–positive and not taking ART, 1.5% tested HIV– positive. In South Africa: Overall, 12.9% self-reported they were HIV–positive and taking ART, 1.3% self-reported they were HIV–positive and not taking ART, 1.9% tested HIV–positive.

<sup>2</sup> For subgroup analysis of males and females, adjusted for triplet and estimating age-group-specific odds ratios to compare HIV–positive with HIV–negative individuals (interaction between age group and HIV status, in the logistic regression model); for analysis of males and females combined, adjusted for triplet and allowing the association between age group and prevalent TB to be different for males and females (interaction between sex and age group, in the logistic regression model). <sup>3</sup> Number of individuals with prevalent TB.

<sup>4</sup> Total participants.

<sup>5</sup> Percentage with prevalent TB.

<sup>6</sup> P-value for evidence for variation in ORs (comparing HIV–positive to HIV–negative individuals) among the 4 age groups.

ART, antiretroviral therapy; CI, confidence interval; HIV, human immunodeficiency virus; OR, odds ratio.

https://doi.org/10.1371/journal.pmed.1004278.t003

odds ratios were 1.40 [95% CI 0.85, 2.29] among males and 1.93 [95% CI 1.14, 3.25] among females, with no evidence that the odds ratios differed between males and females (p = 0.45). Overall, combining the data on males and females, the age-sex-triplet adjusted odds ratio comparing HIV–positive to HIV–negative individuals was 2.29 [95% CI 1.54, 3.41] in Zambian communities and 1.61 [95% CI 1.13, 2.30] in South African communities. The adjusted odds ratios, comparing HIV–positive with HIV–negative individuals, were similar when restricting HIV–positive individuals to those who self-reported they were HIV–positive and taking ART (S4 Table).

## Sensitivity analyses

Findings about trial arm comparisons were similar in sensitivity analyses (S5 Table), for example in the analysis in which all culture-eligible individuals were assumed to have prevalent TB the adjusted prevalence ratio comparing arms A and B combined with arm C was 1.17 [95% CI 0.68, 2.02], and in the analysis in which all sputum-eligible individuals with missing S1 or S2 Xpert-Ultra test results were assumed not to have prevalent TB the adjusted prevalence ratio comparing arms A and B combined with arm C was 1.17 [95% CI 0.66, 1.89].

## Discussion

## Key findings

We found no evidence that the PopART intervention reduced TB prevalence through UTT for HIV and systematic symptom screening for TB, after 4 years of intervention delivery (2014 to 2017 inclusive) and with TB prevalence measured 2 to 4 years later in 2019 to 2021. This lack of evidence was found overall and also in subgroup analyses among men, women, older and younger individuals, HIV-negative and HIV-positive individuals, and among individuals who had been resident in the study communities for  $\geq$ 5 years. Findings were also similar in sensitivity analyses that made different assumptions about TB prevalence among survey participants with missing data on whether or not they had TB on the day of the survey. Our findings were thus consistent across subgroups of the population and robust to alternative ways of accounting for missing data on prevalent TB. TB prevalence was on average high across the 21 study communities, at around 0.9% overall, around 1.6% averaging across South African communities, and around 0.5% averaging across Zambian communities. TB prevalence was also higher among men than women, among older than younger individuals, and around 2 times higher among HIV-positive compared with HIV-negative individuals. Most individuals who had prevalent TB were eligible to provide sputum samples for diagnostic testing only on the basis of an abnormal X-ray and not based on the TB symptoms they reported.

## Interpretation of findings and consistency with other studies

Several factors could have contributed to the lack of evidence for impact of the PopART intervention on TB prevalence. First, the TB-specific component of the PopART intervention consisted of systematic TB screening based on TB symptoms. National and subnational TB prevalence surveys have consistently found that a considerable proportion of individuals who have prevalent TB on the day of the survey are identified for diagnostic testing only on the basis of an abnormal X-ray and not based on the TB suggestive symptoms they report [15,16]. This was also found in the TREATS TB prevalence survey, in which only around 30% (40% in Zambian communities and 20% in South African communities) of individuals with prevalent TB screened positive on TB symptoms. In a national TB prevalence survey in Zambia in 2013 to 2014 [17], 61% of prevalent TB cases were eligible to provide sputum samples based on the TB symptoms they reported, with less strict criteria to screen positive on TB symptoms

compared with the TREATS survey (at least one of cough  $\geq 2$  weeks, fever  $\geq 2$  weeks, chest pains  $\geq 2$  weeks). In a national TB prevalence survey in South Africa in 2017 to 2019 [18], 42% of prevalent TB cases were eligible to provide sputum samples based on the TB symptoms they reported, again with less strict criteria than in the TREATS survey (at least one of cough of any duration, night sweats for  $\geq 2$  weeks, unexplained weight loss for  $\geq 2$  weeks, unexplained fever for  $\geq 2$  weeks). Evidence is emerging that "subclinical" TB, i.e., persons not experiencing the classical TB symptoms, form a large part of the reservoir of individuals with active TB [19–21].

Second, an interim review of the TB symptom screening component of the PopART intervention identified that it was not prioritised by the CHiPs during years 1 to 2 of intervention delivery. Instead, CHiPs focused on offering universal HIV testing and supporting HIV-positive individuals to register for HIV care, because the primary goal of the intervention was to reduce HIV incidence. CHiPs gave more attention to TB symptom screening during years 3 to 4 of intervention delivery, following re-training and enhanced monitoring, but this effort may have been for too short a time to have much impact on population-wide TB incidence and prevalence. Nevertheless, among individuals who participated in the PopART intervention, the percentage who were eligible to provide sputum samples for TB diagnostic testing based on the TB symptoms they reported was around 1%, 1%, and 2%, respectively, across rounds 1 to 3 of intervention delivery in Zambian communities, the proportion who were diagnosed with TB (among all who participated in the intervention) was around 0.08%, 0.08%, and 0.1%, respectively, and in each round approximately 200,000 individuals aged  $\geq$ 15 years participated. In the third round of intervention delivery in South African communities, the corresponding figures were that around 6% of individuals were eligible to provide sputum samples for TB diagnostic testing based on the TB symptoms they reported, the proportion of all intervention participants who were diagnosed with TB was 0.4%, and around 85,000 individuals aged >15 years participated (estimates for rounds 1 to 2 could not be made due to challenges with some aspects of data collection and recording) [22,23].

Third, we found that TB prevalence was much higher among men than women in the TREATS study communities, consistent with findings from national TB prevalence surveys in Zambia and South Africa [17,18]. However, the PopART intervention was less effective at reaching men than women, for example, in Zambian communities around two-thirds of men participated in the third round of intervention delivery compared with around 85% of women [24]. This lower coverage of the intervention among men is likely a contributing factor to the lack of overall impact of the intervention on community-wide TB prevalence.

Fourth, by the time of the TREATS TB prevalence survey in 2019 to 2021, HIV testing and ART coverage among HIV-positive individuals was on average similar in communities in the standardof-care arm (arm C) to those in the PopART intervention arm (arms A and B). In contrast, between mid-2016 and mid-2018, the proportion of HIV-positive individuals who were virally suppressed was on average lower in the standard-of-care arm than in the PopART intervention arms [7]. Additionally, a UTT intervention may have a smaller effect on TB prevalence than on TB incidence. HIV infection, in the absence of ART, has previously been shown to increase TB prevalence to a lesser extent than TB incidence [25]. This is due to the shorter average duration of TB in PLWH compared with HIV-negative individuals [26], because symptoms progress more rapidly and so TB is identified earlier or individuals may die from TB more quickly [25,27]. It is less well understood how ART may affect this relationship. For example, an individual who is HIV-positive and on ART, while still more likely to have TB than someone who is HIV-negative, may have a disease course that is more like that of an HIV-negative person in terms of the time between disease onset and diagnosis. On the other hand, HIV-positive individuals who are in HIV care are expected to be screened for TB (based on TB symptoms) every time they attend a health clinic for HIV care, which facilitates early diagnosis of TB disease compared with HIV-positive individuals who are not in HIV care.

Fifth, the COVID pandemic disrupted access to TB diagnostic and treatment services during 2020 to 2021, and it also interrupted and delayed the completion of the TB prevalence survey. While this happened across all study communities, it may have reduced the differences among trial arms in terms of TB prevalence.

Our findings are consistent with the TREATS co-primary endpoint of the incidence of new infection with *M. tuberculosis* among young people during 2018 to 2021, for which there was no evidence of lower incidence in arm A compared with arm C communities [28].

## Generalisability, strengths, and limitations

The TREATS TB prevalence survey was large, including around 50,000 individuals, and covering 21 communities. Nevertheless, the study was only powered to detect fairly large reductions (of the order of 40% to 50%) in TB prevalence in the PopART intervention trial arms (arms A and B) compared with the standard-of-care arm, and also the between-community variation in TB prevalence was larger than had been anticipated. Also, the HPTN 071 (PopART) study was designed to be balanced across trial arms on pre-trial HIV prevalence, but not on indicators of TB burden, and for most of the South African communities information on pre-trial TB prevalence was not available. The proportion of individuals who participated in the survey, among all who were invited to participate, was around 5% higher in arms A and B combined compared with arm C, but the age-sex distribution and HIV prevalence among participants was similar across the trial arms, showing that the trial had good balance across trial arms on these key individual characteristics. The missing data on prevalent TB status introduced additional uncertainty to our prevalence estimates and trial arm comparisons, but imputation of missing data was done to limit bias and sensitivity analyses showed that our findings were robust to different assumptions. Lastly, the CHiPs were a similar cadre of staff as could be employed programmatically, so our study provides evidence about the effectiveness of the PopART intervention in reducing TB prevalence in close-to-programmatic conditions.

## Implications of study findings

Our findings indicate that systematic screening for TB that is based on symptom screening alone is not sufficient to achieve a large reduction in TB prevalence over a period of several years. Including chest X-ray screening alongside TB symptom screening could substantially increase the sensitivity of systematic screening for TB and is increasingly feasible with the advent of mobile X-ray machines (including ones that can be transported in a backpack) and computer-assisted diagnosis of digital X-ray images. In settings with high general population TB prevalence (such as the communities in our study), systematic screening for TB based on both symptoms and chest X-ray might be able to achieve a considerable reduction in TB prevalence and transmission, in absolute as well as relative terms; such a strategy could be evaluated in future research. The accuracy, in particular the specificity, of computer-aided diagnosis has been improved since the TREATS TB prevalence survey was conducted [29,30], bringing down the overall costs of X-ray screening by reducing the proportion of individuals who screen positive and are then eligible for diagnostic testing for TB. Additionally, it might be possible to increase the sensitivity of systematic TB screening based on TB symptoms alone through better symptoms enquiry (for example, taking more time over such enquiry to facilitate elicitation of milder or less obvious symptoms).

The 2021 WHO TB screening guidelines included a statement that "Systematic screening for TB disease may be conducted among the general population in areas with an estimated TB prevalence of 0.5% or higher," but the evidence for this is weak. They also stated that a range of approaches to such screening could be considered, including diagnostic testing (using

molecular tests) for everyone or for those who screen positive on TB symptoms or chest X-ray. Our study provides rigorous evidence about the impact of systematic TB symptom screening on TB prevalence in communities with an average prevalence of >0.5%, which can be taken into account in future updates to these guidelines.

Our study does not provide clear evidence about the effect of UTT for HIV on TB prevalence, because ART coverage was high in all trial arms by the time of the TREATS TB prevalence survey. However, TB prevalence remained approximately 2 times higher among HIV– positive than HIV–negative individuals, after adjusting for sex and age group and study triplet, compared to around 3 to 4 times higher in similar study communities in the ZAMSTAR trial TB prevalence survey that was conducted in 2010 when ART coverage was much lower [31]. This suggests that the overall impact of increased coverage of HIV care and ART is to reduce TB prevalence among HIV–positive individuals.

## Conclusions

There was no evidence that the PopART intervention reduced TB prevalence. Potential explanations include: similarly high coverage of HIV testing and treatment in all trial arms by 2019 to 2021; systematic TB symptom screening missing subclinical TB; ART coverage affecting TB prevalence less than TB incidence; the initially lower priority given by the CHiPs to the TB screening component of the PopART intervention; lower coverage of the PopART intervention among men than women; and COVID-19 pandemic effects on TB prevalence during 2020 to 2021. Our findings indicate that the overall impact of increased coverage of HIV care and ART is to reduce TB prevalence among HIV–positive individuals.

## Supporting information

**S1 Fig. Sputum eligibility definition.** (TIF)

S2 Fig. Culture eligibility algorithm based on the results of Xpert-Ultra testing on 2 sputum samples (S1 and S2).

(TIF)

**S3 Fig. Decision tree for culture eligibility and collection of S3 and S4 samples.** (TIF)

S4 Fig. Algorithm to determine final culture result based on the culture results from each of 2 MGIT tubes.

(TIF)

S5 Fig. Flow diagram by country of participant enrolment, sample collection, and MTB detection.

(TIF)

S6 Fig. Flow diagram by 3 trial arms of participant enrolment, sample collection, and MTB detection.

(TIF)

**S1 Text. Culture testing of sputum samples.** (DOCX)

**S2 Text. Details on missing value imputation.** (DOCX)

S1 Table. Characteristics of the TREATS study participants population by arm, country, and overall totals.

(DOCX)

**S2** Table. HIV indicators collected during TREATS by arm, country, and overall totals. (DOCX)

**S3** Table. Characteristics of participants with prevalent TB, by trial arm and country. (DOCX)

S4 Table. TB prevalence by country, comparing individuals who self-reported they were HIV-positive and taking ART with HIV-negative individuals, overall and stratified by sex and age group.

(DOCX)

**S5 Table. TB prevalence by trial arm, sensitivity analyses.** (DOCX)

**S1 Consort Checklist. CONSORT 2010 checklist of information to include when reporting a cluster-randomised trial.** Extension of CONSORT for abstracts to reports of cluster-randomised trials.

(PDF)

**S1** Protocol. Tuberculosis Reduction through Expanded Anti-Retroviral Treatment and Screening (TREATS) Project. (PDF)

**S1 SAP. Statistical analysis plan.** (PDF)

## Acknowledgments

The authors would like to express their appreciation and gratitude to all members of the TREATS study team in Zambia and South Africa for their hard work in sometimes difficult circumstances. We also appreciate the participation, time, and information sharing of the study participants and their communities. We are also grateful to all members of the HPTN 071 (PopART) Study Team for carrying out the intervention with dedication and commitment and to the PopART study participants and their communities for their contributions to the research. The following persons are additional members of the TREATS study team that are not co-authors on this manuscript: Alwyn Mwinga, Virginia Bond, Musonda Simwinga, Tila Mainga, Modupe Amofa-Sekyi, Maina Cheeba, Bxyn Kangololo, Beatrice Nyondo, Robynn Paulsen, Carmen Sisam, Algernon Africa, Lily Telisinghe, Ranjeeta Thomas, Frank Vijn, and Vladyslav Nikolayevskyy. The content of this paper is solely the responsibility of the authors.

## **Author Contributions**

**Conceptualization:** Eveline Klinkenberg, Sian Floyd, Kwame Shanaube, Petra de Haas, Barry Kosloff, Peter J. Dodd, Nico Kalisvaart, Albertus Schaap, Sarah Fidler, Richard Hayes, Helen Ayles.

Data curation: Thomas Gachie, Nico Kalisvaart, Nkatya Kasese, Albertus Schaap.

Formal analysis: Eveline Klinkenberg, Sian Floyd.

Funding acquisition: Kwame Shanaube, Richard Hayes, Helen Ayles.

Investigation: Eveline Klinkenberg.

- **Methodology:** Eveline Klinkenberg, Sian Floyd, Kwame Shanaube, Thomas Gachie, Petra de Haas, Barry Kosloff, Peter J. Dodd, Nico Kalisvaart, Nkatya Kasese, Redwaan Vermaak, Albertus Schaap, Richard Hayes, Helen Ayles.
- **Project administration:** Kwame Shanaube, Linda Mureithi, Maria Ruperez, Chali Wapamesa, James Michael Burnett, Helen Ayles.
- Resources: Linda Mureithi, Helen Ayles.
- **Software:** Thomas Gachie, Nico Kalisvaart, Nkatya Kasese, Redwaan Vermaak, Albertus Schaap.
- **Supervision:** Eveline Klinkenberg, Kwame Shanaube, Linda Mureithi, Thomas Gachie, Petra de Haas, Barry Kosloff, Maria Ruperez, Chali Wapamesa, James Michael Burnett, Nico Kalisvaart, Nkatya Kasese, Redwaan Vermaak, Albertus Schaap, Helen Ayles.
- Validation: Eveline Klinkenberg, Sian Floyd, Thomas Gachie, Petra de Haas, Nico Kalisvaart, Nkatya Kasese, Redwaan Vermaak, Albertus Schaap, Helen Ayles.
- Visualization: Thomas Gachie.

Writing – original draft: Eveline Klinkenberg, Sian Floyd.

Writing – review & editing: Eveline Klinkenberg, Sian Floyd, Kwame Shanaube, Linda Mureithi, Thomas Gachie, Petra de Haas, Barry Kosloff, Peter J. Dodd, Maria Ruperez, Nkatya Kasese, Albertus Schaap, Sarah Fidler, Richard Hayes, Helen Ayles.

#### References

- 1. WHO. The End TB strategy. Geneva: World Health Organization; 2015. Contract No.: WHO/HTM/TB/ 2015.19. Available from: https://apps.who.int/iris/rest/bitstreams/1271371/retrieve; access date 2023 May 26.
- WHO. Consolidated guidelines on tuberculosis. Module 2: screening–systematic screening for tuberculosis disease. Geneva: World Health Organization; 2021. Report No.: Licence: CC BY-NC-SA 3.0 IGO. Contract No.: ISBN: 978–92–4–002267–6. Available from: https://www.who.int/publications/i/item/9789240022676; access date 2023 May 26.
- Marks GB, Nguyen NV, Nguyen PTB, Nguyen TA, Nguyen HB, Tran KH, et al. Community-wide Screening for Tuberculosis in a High-Prevalence Setting. N Engl J Med. 2019; 381(14):1347–1357. https://doi.org/10.1056/NEJMoa1902129 PMID: 31577876
- WHO. Guidelines for intensified tuberculosis case-finding and isoniazid preventive therapy for people living with HIV in resource constrained settings. Geneva: World Health Organization; 2011. Report No.: ISBN 978 92 4 150070 8. Available from: https://apps.who.int/iris/bitstream/handle/10665/44472/ 9789241500708\_eng.pdf; access date 2023 May 26.
- WHO. Consolidated guidelines on the use of antiretroviral drugs for treating and preventing HIV infection: recommendations for a public health approach. Geneva: World Health Organization; 2016. Report No.: ISBN 978 92 4 154968 4. Available from: https://www.who.int/publications/i/item/9789241549684 access date 26 May 2023
- Havlir D, Lockman S, Ayles H, Larmarange J, Chamie G, Gaolathe T, et al. What do the Universal Test and Treat trials tell us about the path to HIV epidemic control? J Int AIDS Soc. 2020; 23(2):e25455. https://doi.org/10.1002/jia2.25455 PMID: 32091179
- Hayes RJ, Donnell D, Floyd S, Mandla N, Bwalya J, Sabapathy K, et al. Effect of Universal Testing and Treatment on HIV Incidence—HPTN 071 (PopART). N Engl J Med. 2019; 381(3):207–218.
- Bock P, Jennings K, Vermaak R, Cox H, Meintjes G, Fatti G, et al. Incidence of Tuberculosis Among HIV-Positive Individuals Initiating Antiretroviral Treatment at Higher CD4 Counts in the HPTN 071 (PopART) Trial in South Africa. J Acquir Immune Defic Syndr. 2018; 77(1):93–101. <u>https://doi.org/10.1097/QAI.00000000001560 PMID: 29016524</u>
- Hayes R, Ayles H, Beyers N, Sabapathy K, Floyd S, Shanaube K, et al. HPTN 071 (PopART): rationale and design of a cluster-randomised trial of the population impact of an HIV combination prevention

intervention including universal testing and treatment—a study protocol for a cluster randomised trial. Trials. 2014; 15:57. https://doi.org/10.1186/1745-6215-15-57 PMID: 24524229

- Ayles H, Muyoyeta M, Du Toit E, Schaap A, Floyd S, Simwinga M, et al. Effect of household and community interventions on the burden of tuberculosis in southern Africa: the ZAMSTAR community-randomised trial. Lancet. 2013; 382(9899):1183–1194. https://doi.org/10.1016/S0140-6736(13)61131-9 PMID: 23915882
- Claassens MM, van Schalkwyk C, Floyd S, Ayles H, Beyers N. Symptom screening rules to identify active pulmonary tuberculosis: Findings from the Zambian South African Tuberculosis and HIV/AIDS Reduction (ZAMSTAR) trial prevalence surveys. PLoS ONE. 2017; 12(3):e0172881. https://doi.org/10. 1371/journal.pone.0172881 PMID: 28257424
- 12. Cepheid. Xpert MTB/RIF Ultra; GXMTB/RIF-ULTRA-10 & GXMTB/RIF-ULTRA-50. In: Cepheid, editor. package insert ed. Sweden: Cepheid; 2019.
- Floyd S, Klinkenberg E, de Haas P, Kosloff B, Gachie T, Dodd PJ, et al. Optimising Xpert-Ultra and culture testing to reliably measure tuberculosis prevalence in the community: findings from surveys in Zambia and South Africa. BMJ Open. 2022; 12(6):e058195. https://doi.org/10.1136/bmjopen-2021-058195 PMID: 35710250
- 14. Hayes R, Moulton LH. Cluster Randomised Trials. Chapman & Hall; 2017, June 13, 2022. p. 424.
- Law I, Floyd K. National tuberculosis prevalence surveys in Africa, 2008–2016: an overview of results and lessons learned. Trop Med Int Health. 2020; 25(11):1308–1327. https://doi.org/10.1111/tmi.13485 PMID: 32910557
- Onozaki I, Law I, Sismanidis C, Zignol M, Glaziou P, Floyd K. National tuberculosis prevalence surveys in Asia, 1990–2012: an overview of results and lessons learned. Trop Med Int Health. 2015; 20 (9):1128–1145. https://doi.org/10.1111/tmi.12534 PMID: 25943163
- Kapata N, Chanda-Kapata P, Ngosa W, Metitiri M, Klinkenberg E, Kalisvaart N, et al. The Prevalence of Tuberculosis in Zambia: Results from the First National TB Prevalence Survey, 2013–2014. PLoS ONE. 2016; 11(1):e0146392. https://doi.org/10.1371/journal.pone.0146392 PMID: 26771588
- Moyo S, Ismail F, Van der Walt M, Ismail N, Mkhondo N, Dlamini S, et al. Prevalence of bacteriologically confirmed pulmonary tuberculosis in South Africa, 2017–19: a multistage, cluster-based, cross-sectional survey. Lancet Infect Dis. 2022; 22(8):1172–1180. <u>https://doi.org/10.1016/S1473-3099(22)</u> 00149-9 PMID: 35594897
- Bajema KL, Bassett IV, Coleman SM, Ross D, Freedberg KA, Wald A, et al. Subclinical tuberculosis among adults with HIV: clinical features and outcomes in a South African cohort. BMC Infect Dis. 2019; 19(1):14. https://doi.org/10.1186/s12879-018-3614-7 PMID: 30611192
- Kendall EA, Shrestha S, Dowdy DW. The Epidemiological Importance of Subclinical Tuberculosis. A Critical Reappraisal. Am J Respir Crit Care Med. 2021; 203(2):168–174. <u>https://doi.org/10.1164/rccm.</u> 202006-2394PP PMID: 33197210
- Stuck L, van Haaster AC, Kapata-Chanda P, Klinkenberg E, Kapata N, Cobelens F. How "Subclinical" is Subclinical Tuberculosis? An Analysis of National Prevalence Survey Data from Zambia. Clin Infect Dis. 2022; 75(5):842–8. https://doi.org/10.1093/cid/ciab1050 PMID: 34984431
- Gachie T, Team HPS. Outcomes of household-based, community TB case finding from the HPTN 071 (PopART) study in Zambia. 50th Union World Conference on Lung Health; 1 November 2019; Hyderabad, India 2019.
- 23. Gachie T, Team HPS. What can we learn from a combined TB screening, HIV testing and treatment door-to-door community-wide strategy, repeated annually (for a period of 3 years) in settings with a high burden of TB/HIV? New horizons for medical statistics & epidemiology in global health A symposium celebrating the 50th anniversary of the International Statistics and Epidemiology Group—June 28-30th 2022 virtual. 2022.
- 24. Floyd S, Shanaube K, Yang B, Schaap A, Griffith S, Phiri M, et al. HIV testing and treatment coverage achieved after 4 years across 14 urban and peri-urban communities in Zambia and South Africa: An analysis of findings from the HPTN 071 (PopART) trial. PLoS Med. 2020; 17(4):e1003067. https://doi.org/10.1371/journal.pmed.1003067 PMID: 32240156
- Corbett EL, Watt CJ, Walker N, Maher D, Williams BG, Raviglione MC, et al. The growing burden of tuberculosis: global trends and interactions with the HIV epidemic. Arch Intern Med. 2003; 163 (9):1009–1021. https://doi.org/10.1001/archinte.163.9.1009 PMID: 12742798
- Ku CC, MacPherson P, Khundi M, Nzawa Soko RH, Feasey HRA, Nliwasa M, et al. Durations of asymptomatic, symptomatic, and care-seeking phases of tuberculosis disease with a Bayesian analysis of prevalence survey and notification data. BMC Med. 2021; 19(1):298. https://doi.org/10.1186/s12916-021-02128-9 PMID: 34753468

- Dimairo M, MacPherson P, Bandason T, Zezai A, Munyati SS, Butterworth AE, et al. The risk and timing of tuberculosis diagnosed in smear-negative TB suspects: a 12 month cohort study in Harare, Zimbabwe. PLoS ONE. 2010; 5(7):e11849. https://doi.org/10.1371/journal.pone.0011849 PMID: 20676374
- 28. Shanaube K, Schaap A, Mureithi L, Amofa-Sekyi M, Paulsen R, Cheeba M, et al. The impact of a combined TB/HIV intervention on the incidence of TB infection in the HPTN 071 (PopART) trial communities in Zambia and South Africa. PLoS Glob Public Health. 2023. in press.
- Murphy K, Habib SS, Zaidi SMA, Khowaja S, Khan A, Melendez J, et al. Computer aided detection of tuberculosis on chest radiographs: An evaluation of the CAD4TB v6 system. Sci Rep. 2020; 10 (1):5492. https://doi.org/10.1038/s41598-020-62148-y PMID: 32218458
- Qin ZZ, Ahmed S, Sarker MS, Paul K, Adel ASS, Naheyan T, et al. Tuberculosis detection from chest xrays for triaging in a high tuberculosis-burden setting: an evaluation of five artificial intelligence algorithms. Lancet Digit Health. 2021; 3(9):e543–e554. https://doi.org/10.1016/S2589-7500(21)00116-3 PMID: 34446265
- Yates TA, Ayles H, Leacy FP, Schaap A, Boccia D, Beyers N, et al. Socio-economic gradients in prevalent tuberculosis in Zambia and the Western Cape of South Africa. Trop Med Int Health. 2018; 23 (4):375–390. https://doi.org/10.1111/tmi.13038 PMID: 29432669